Author’s response to reviews

Title: Exploring effectiveness of different health financing mechanisms in Nigeria; what needs to change and how can it happen?

Authors:

Obinna Onwujekwe (onwujekwe@yahoo.co.uk)
Nkoli Ezumah (ezumahnk@yahoo.com)
Chinyere Mbachi (chinyere23ng@gmail.com)
Felix Obi (obifelix@gmail.com)
Hyacinth Ichoku (hichoku@yahoo.com)
Benjamin Uzochukwu (bscuzochukwu@gmail.com)
Hong Wang (Hong.Wang@gatesfoundation.org)

Version: 1 Date: 15 Apr 2019

Author’s response to reviews:

1. Overlap

We note that the current submission contains some textual overlap with other previously published works

RESPONSE: The textual overlaps have been rephrased and revised.

2. Title Page

Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

RESPONSE: Email addresses for all authors have been included on the title page

3. Manuscript Headings

In order to be in line with journal requirements please change your “Study Methods” heading to instead read “Methods”.
RESPONSE: This has been changed.

4. Funding

In the “Funding” section in the Declarations, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

RESPONSE: The role of the funding in study design, and data collection, analysis and interpretation has been elaborated.

5. Clean Manuscript

RESPONSE: A clean manuscript has been uploaded.